Author’s response to reviews

Title: Combining Wireless technology and Behavioral Economics to Engage Patients with cardiometabolic disease (WiBEEP): A Pilot Study

Authors:

Edith Angellotti (EAngellotti@tuftsmedicalcenter.org)

John Wong (JWong@tuftsmedicalcenter.org)

Ayal Pierce (Ayal.Pierce@tufts.edu)

Benjamin Hescott (b.hescott@northeastern.edu)

Anastassios Pittas (APittas@tuftsmedicalcenter.org)

Version: 2 Date: 21 Dec 2018

Author’s response to reviews:

We would like to thank the Reviewers for their valuable comments, which we considered in improving the revised manuscript. Changes in the text are indicated by the track-changes mode and a clean copy is also included.

Response to Specific Comments by Reviewer 1

1. Line 194-195: Please provide a reference for this statement.

Response: As suggested, we provided a reference.

2. Line 195: Consider changing to "Recent evidence supports the…"

Response: We rephrased the sentence (lines 194-196).